Author's response to reviews

Title: Cross-sectional study of hepatitis B virus infection in rural communities, Central African Republic

Authors:

Narcisse P Komas (npkomas@yahoo.fr)
Ulrich Vickos (vickos.ulrich@gmx.com)
Judith M Hübschen (judith.huebschen@crp-sante.lu)
Aubin Béré (berostau@yahoo.fr)
Alexandre Manirakiza (amanirak@yahoo.fr)
Claude P Muller (Claude.Muller@crp-sante.lu)
Alain Le Faou (alainlefaou@univ-lorraine.fr)

Version: 4 Date: 23 May 2013

Author's response to reviews:

Dear Editor,

Thank you for your remarks about our revised manuscript, MS: 1593046211922741. See enclosed our response to your remarks:

1- The sample was not random and the method of recruitment needs to be described.
Response: Method of recruitment is described (see in bold in manuscript in study population section)

2- Can the authors confirm that age and sex standardised, given the sample was not representative of the sex distribution of the population
Response: Age and sex are standardised using the CAR population distribution according to CAR 2004 census (see in bold in manuscript in results section and value is in Table 1)